Author’s response to reviews

Title: Development of an educational intervention for patients with Irritable Bowel Syndrome (IBS) - a pilot study.

Authors:

Gisela Ringstrom (gisela.ringstrom@vgregion.se)
Stine Storsrud (stine.storsrud@vgregion.se)
Sara Lundqvist (sara.lundqvist@vgregion.se)
Berndt Westman (berndt.westman@privat.utfors.se)
Magnus Simren (magnus.simren@medicine.gu.se)

Version: 5 Date: 2 February 2009

Author’s response to reviews:

Regarding MS: 9268565921279939, entitled “Development of an educational intervention for patients with irritable bowel syndrome (IBS) - a pilot study.”

We have now recreated the tables according to your recommendation.

On behalf of all the co-authors,

Gisela Ringström and Magnus Simrén